Author's response to reviews

Title: Small intestinal bacterial overgrowth in irritable bowel syndrome: are there any predictors?

Authors:

    Savio C Reddymasu (sreddymasu@kumc.edu)
    Sandra Sostarich (ssostra@kumc.edu)
    Richard W McCallum (rmccallu@Kumc.edu)

Version: 7 Date: 18 February 2010

Author's response to reviews: see over
Kansas City, KS
USA
February 18, 2010
To
The Editor,
BMC Gastroenterology.

Re: Submission of revisions to MS: 8317536724676099- Small intestinal bacterial overgrowth in irritable bowel syndrome: are there any predictors?

We have made the necessary formatting changes requested by the editorial team.

We would like to thank you and the reviewers again for their input into our manuscript. We have tried to revise the manuscript by incorporating their suggestions and believe that this strengthened the manuscript and hope that the current version of the article is more acceptable. However if there is still thought to be any shortcomings in the article, we would be willing to make further revisions as felt appropriate by the Editor and reviewers.

Regards,

Savio C Reddymasu, MD
and
Richard W McCallum, MD